Author's response to reviews

Title: The TURN-OUT Trial: Transverse position. Using Rotation to aid Normal birth: OUTcomes following manual rotation.

Authors:

Bradley N/A de Vries Dr (bradley.devries@sswahs.nsw.gov.au)
Hala N/A Phipps Mrs (hala.phipps@sswahs.nsw.gov.au)
Sabrina N/A kuah Dr (sabrina.Kuah@health.sa.gov.au)
John N/A Pardey Dr (john@pardey.com.au)
Joanne N/A Ludlow Dr (Joanne.ludlow@sswahs.nsw.gov.au)
Andrew N/A Bisits Assoc Prof
(Andrew.Bisits@SESIAHS.HEALTH.NSW.GOV.AU)
Felicity N/A Park Dr (felicit@med.usyd.edu.au)
David N/A Kowalski Dr (dkowalski@optusnet.com.au)
Jon A Hyett Prof (Jon.hyett@sswahs.nsw.gov.au)

Version: 2
Date: 18 September 2014

Author's response to reviews:

Dear Jessie

Apologies for the delay. The Protocol has now been formatted to meet the Journal's standard. The Title page has been amended to have the required information re authors.

Much appreciated

Hala Phipps